 Vc/21-Feb-05
Week 08 starting 21 Feb 05
SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 21 feb 05 @ 2/21/2005 3:21:51 PM   
	8
	A1
	MBBR3291
	ICS 4 Final connection
	291.87

	4
	A2
	MBBR3269
	Prep 5 Pumping + Cool Down
	1.93 HV

	2
	B1
	SSS110
	PTE 3 Correctors Powering
	1.90

	5
	B2
	MBAR3288
	PT 13.1 IAP @ Warm
	299.11

	10
	C1
	[image: image1.png]



	[image: image2.png]



	291.87

	1
	C2
	MBAL1080
	Prep 5 Pumping + Cool Down
	81.71

	6
	D1
	SSS073
	Prep 2.3 Res. Meas. CDW1
	91.97

	9
	D2
	[image: image3.png]



	Current Lead repair
	292.94

	3
	E1
	MBAL3171
	PT 12 Warm Up
	283.48

	11
	E2
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	295.18

	12
	F1
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	293.57

	7
	F2
	MBBL3153
	ICS 4 Final connection
	291.78


2/21/2005 3:22:54 PM    Temperature data are taken from TBSM18 site 
Coordinator this week: Nikolay; MTM support on call :   16-3602 .

14  Magnets Tested   ( not counting sss108) 
10  dipoles( 2 with MM)+ 4  SSS(0  with MM) +  0  SSSS 
    Disconnected after tests last week ( Sat Midnight to Sat midnight) 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· Situation on Repair :
· Silver Shoes bench F etc
General Points 
e-Traveller & Paper Traveller

  Power Converters  Slow PA & filter by AB-PO 
AOB

	Magnets under Test from 13-02-2005 to 19-02-2005 

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA1
	MBAL2123 
	[image: image8.png]



	04-02-2005 
	50 h.
	30 h.
	142 h.
	14 h.
	22 h.
	15-02-2005 
	264 hours
	
	

	2
	TBA2
	MBAR3278 
	[image: image9.png]



	13-02-2005 
	26 h.
	35 h.
	47 h.
	16 h.
	9 h.
	18-02-2005 
	120 hours
	
	

	3
	TBB1
	SSS034 
	[image: image10.png]



	10-02-2005 
	36 h.
	19 h.
	30 h.
	7 h.
	29 h.
	16-02-2005 
	144 hours
	
	

	4
	TBB1
	SSS108 
	[image: image11.png]



	17-02-2005 
	23 h.
	-
	-
	-
	-
	18-02-2005 
	24 hours
	
	

	5
	TBB2
	MBAL3165 
	[image: image12.png]



	11-02-2005 
	39 h.
	39 h.
	30 h.
	16 h.
	13 h.
	17-02-2005 
	144 hours
	
	

	6
	TBC1
	MBAR3286 
	[image: image13.png]



	08-02-2005 
	36 h.
	29 h.
	56 h.
	13 h.
	13 h.
	14-02-2005 
	144 hours
	
	

	7
	TBC2
	MBAL3168 
	[image: image14.png]



	14-02-2005 
	32 h.
	21 h.
	31 h.
	18 h.
	22 h.
	19-02-2005 
	120 hours
	
	

	8
	TBC2
	MBAR3253 
	[image: image15.png]



	08-02-2005 
	23 h.
	36 h.
	49 h.
	13 h.
	19 h.
	14-02-2005 
	144 hours
	
	

	9
	TBD1
	SSS003 
	[image: image16.png]



	14-02-2005 
	46 h.
	13 h.
	24 h.
	14 h.
	33 h.
	19-02-2005 
	120 hours
	
	

	10
	TBD1
	SSS032 
	[image: image17.png]



	09-02-2005 
	43 h.
	14 h.
	25 h.
	9 h.
	20 h.
	14-02-2005 
	120 hours
	
	

	11
	TBD2
	SSS107 
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	08-02-2005 
	87 h.
	22 h.
	16 h.
	8 h.
	27 h.
	15-02-2005 
	168 hours
	
	

	12
	TBE1
	MBAL3141 
	[image: image19.png]



	12-02-2005 
	25 h.
	38 h.
	33 h.
	13 h.
	13 h.
	17-02-2005 
	120 hours
	
	

	13
	TBE2
	MBBL2146 
	[image: image20.png]



	10-02-2005 
	35 h.
	34 h.
	25 h.
	19 h.
	10 h.
	16-02-2005 
	144 hours
	
	

	14
	TBF1
	MBAR3285 
	[image: image21.png]



	10-02-2005 
	88 h.
	21 h.
	25 h.
	16 h.
	10 h.
	17-02-2005 
	168 hours
	
	

	15
	TBF2
	MBBR3270 
	[image: image22.png]



	10-02-2005 
	32 h.
	27 h.
	32 h.
	21 h.
	6 h.
	15-02-2005 
	120 hours
	
	


Quench Information
 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL2123
	11320.6 A.
(D2-U-S13)
	12523.5 A.
(D1-L-S10)
	12675.9 A.
(D1-U-S01)
	12850 A.
[image: image23.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAR3278
	12009 A.
(D2-U)
	12255 A.
(D2-U-S01)
	12419 A.
(D2-U-S09)
	12590.7 A.
(D2-U-S05)
	12689 A.
(D1 - U)
	12850 A.
[image: image24.jpg]



	12850 A.
[image: image25.jpg]



	12850 A.
[image: image26.jpg]



	-
	-
	-
	-
	-
	-
	-

	3
	SSS034
	12281.4 A.
(QE_12)
	12256.3 A.
(QE-12)
	12664 A.
(QE-12)
	12687 A.
(QE-34)
	12850 A.
[image: image27.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	
	

	4
	SSS108
	[image: image28.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	5
	MBAL3165
	10106 A.
(D2_L)
	12303.3 A.
(D2 - L)
	[image: image29.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	6
	MBAR3286
	11801.2 A.
(D2-U)
	12128.9 A.
(D2-U)
	[image: image30.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	7
	MBAL3168
	9734 A.
(D2_L)
	11471 A.
(D2 - U)
	12612 A.
(D2-U)
	12574 A.
(D2-U)
	12755 A.
(D2-U)
	12850 A.
[image: image31.jpg]



	-
	-
	-
	-
	-
	-
	-
	
	

	8
	MBAR3253
	4858 A.
(Vcla int)
	6690 A.
(D2-L)
	11379.6 A.
(D2-L)
	12125 A.
(D2_L)
	12950.7 A.
(D1-L)
	[image: image32.png]



	-
	-
	-
	-
	-
	-
	-
	
	

	9
	SSS003
	11604.3 A.
(Q_I_12)
	12111.5 A.
(No Data)
	[image: image33.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	10
	SSS032
	12818.8 A.
(QI-34)
	12683 A.
(QI-34)
	12850 A.
[image: image34.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	11
	SSS107
	12249.3 A.
(QI_34)
	12848 A.
(QI_34)
	[image: image35.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	12
	MBAL3141
	11514.6 A.
(D2-U)
	12745 A.
(D2_U)
	12850 A.
[image: image36.jpg]



	12856 A.
[image: image37.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	13
	MBBL2146
	10351.4 A.
(D2_U)
	12230.1 A.
(D1-L)
	12727 A.
(D1_L)
	[image: image38.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	14
	MBAR3285
	8391 A.
(D2-L)
	12782 A.
(D2-L)
	[image: image39.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	15
	MBBR3270
	12526 A.
(D1-U-S02)
	12850 A.
[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	

	From 13-02-2005 to 19-02-2005 

	-
	Bench
	Magnet Name
	shafts
	Start Date .
	Connect
Time
	Wait
+Cooling
	Cold
Tests
	Wait 
+Warming
	Disconnect
Time
	End Date .
	Total
	Q

	1
	TBF2
	MBBR3270 
	True 
	10-02-2005 
	32 h.
	27 h.
	32 h.
	21 h.
	6 h.
	15-02-2005 
	5 days
	1 

	2
	TBF1
	MBAR3285 
	False 
	10-02-2005 
	88 h.
	21 h.
	25 h.
	16 h.
	10 h.
	17-02-2005 
	7 days
	2 

	3
	TBE2
	MBBL2146 
	False 
	10-02-2005 
	35 h.
	34 h.
	25 h.
	19 h.
	10 h.
	16-02-2005 
	6 days
	3 

	4
	TBE1
	MBAL3141 
	False 
	12-02-2005 
	25 h.
	38 h.
	33 h.
	13 h.
	13 h.
	17-02-2005 
	5 days
	2 

	5
	TBD2
	SSS107 
	False 
	08-02-2005 
	87 h.
	22 h.
	16 h.
	8 h.
	27 h.
	15-02-2005 
	7 days
	2 

	6
	TBD1
	SSS003 
	False 
	14-02-2005 
	46 h.
	13 h.
	24 h.
	14 h.
	33 h.
	19-02-2005 
	5 days
	2 

	7
	TBD1
	SSS032 
	False 
	09-02-2005 
	43 h.
	14 h.
	25 h.
	9 h.
	20 h.
	14-02-2005 
	5 days
	2 

	8
	TBC2
	MBAL3168 
	False 
	14-02-2005 
	32 h.
	21 h.
	31 h.
	18 h.
	22 h.
	19-02-2005 
	5 days
	5 

	9
	TBC2
	MBAR3253 
	False 
	08-02-2005 
	23 h.
	36 h.
	49 h.
	13 h.
	19 h.
	14-02-2005 
	6 days
	5 

	10
	TBC1
	MBAR3286 
	False 
	08-02-2005 
	36 h.
	29 h.
	56 h.
	13 h.
	13 h.
	14-02-2005 
	6 days
	2 

	11
	TBB2
	MBAL3165 
	False 
	11-02-2005 
	39 h.
	39 h.
	30 h.
	16 h.
	13 h.
	17-02-2005 
	6 days
	2 

	12
	TBB1
	SSS034 
	False 
	10-02-2005 
	36 h.
	19 h.
	30 h.
	7 h.
	29 h.
	16-02-2005 
	6 days
	4 

	13
	TBB1
	SSS108 

HV orbit Cor
	False 
	17-02-2005 
	23 h.
	88 h.
	-87 h.
	1 h.
	8 h.
	18-02-2005 
	1 days
	0 

	14
	TBA2
	MBAR3278 
	True 
	13-02-2005 
	26 h.
	35 h.
	47 h.
	16 h.
	9 h.
	18-02-2005 
	5 days
	5 

	15
	TBA1
	MBAL2123 
	True 
	04-02-2005 
	50 h.
	30 h.
	142 h.
	14 h.
	22 h.
	15-02-2005 
	11 days
	3 


 
Week 14 Feb05 to 20Feb 05
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OP statistics for SM18 OP during the period from 14/02/2005 06:00:00 to 21/02/2005 21:00:00
	Lines 
	in Super Cycle 
	in Faults 
	Availabilities (%) 

	TBA1
	126 [h] 44 [min] 43[s]
	13 [h] 02 [min] 16[s]
	89.71

	TBA2
	126 [h] 44 [min] 43[s]
	11 [h] 17 [min] 25[s]
	91.09

	TBB1
	141 [h] 34 [min] 37[s]
	17 [h] 45 [min] 22[s]
	87.46

	TBB2
	126 [h] 44 [min] 43[s]
	38 [min] 01[s]
	99.5

	TBC1
	126 [h] 44 [min] 43[s]
	38 [min] 01[s]
	99.5

	TBC2
	126 [h] 44 [min] 43[s]
	17 [h] 40 [min] 55[s]
	86.05

	TBD1
	126 [h] 44 [min] 43[s]
	27 [h] 45 [min] 09[s]
	78.1

	TBD2
	126 [h] 44 [min] 43[s]
	1 [h] 06 [min] 08[s]
	99.13

	TBE1
	142 [h] 44 [min] 43[s]
	16 [h] 38 [min] 01[s]
	88.35

	TBE2
	126 [h] 44 [min] 43[s]
	5 [h] 08 [min] 01[s]
	95.95

	TBF1
	126 [h] 44 [min] 43[s]
	3 [h] 34 [min] 32[s]
	97.18

	TBF2
	142 [h] 44 [min] 43[s]
	16 [h] 38 [min] 01[s]
	88.35

	TOTAL: 
	1567 [h] 46 [min] 30[s]
	131 [h] 51 [min] 52[s]
	91.59


	Faults Groups 
	Faults Names 
	Duration 

	Cryogenics
	Miscellaneous
	14 [h] 06 [min] 08[s]

	
	
	

	Electronics
	-----
	12 [h] 46 [min] 49[s]

	
	
	

	HV insulation
	-----
	44 [h] 24 [min] 15[s]

	
	
	

	Hardware
	Power converters
	4 [h] 51 [min] 28[s]

	
	
	

	ICS
	-----
	3 [h] 37 [min] 06[s]

	
	
	

	Miscellaneous
	-----
	21 [h] 40 [min] 53[s]

	
	
	

	Network
	-----
	3 [h] 13 [min] 02[s]

	
	
	

	Software
	MTF
	21 [min] 15[s]

	
	Test master
	1 [h] 32 [min] 53[s]
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Bottom of Form

	[SM18 OP] 17/02/2005 05:47:56

	Still fighting with RMD. AB/PO second line was called. Richard. 

	 

	[SM18 OP] 17/02/2005 04:00:00

	RMD is returning a "PC fault" signal, that we are unable to clear. AB/PO first line was called. Richard. 

	[SM18 OP] 16/02/2005 23:10:00

	Local PLC for water control was found to be in TEMA mode, due to which the valves could not be opened. Contacted Richard and he advised to put in LOCAL mode. Problem solved. 

	 

	[SM18 OP] 15/02/2005 23:50:36

	TBA2: PT1 - IAP @ cold. Dc offset errors in QA compensation (eg. CF_S07_D). Don't know how to adjust. Talked to Richard... he will look into it but told us to proceed futher 

	 

	[SM18 OP] 15/02/2005 19:52:38

	TBD1 &2: Problem with generation of test reports of magnet SSS074 and SSS003 was solved by Marco. It was due to a template file corrupted. Richard. 

	 

	[SM18 OP] 15/02/2005 19:50:53

	TBC1: HV problem on this bench is now understood. The defect comes from the water cooled power cables. Richard. 

	 

	[SM18 OP] 15/02/2005 18:02:48

	ALL: Test reports of HV tests performed during the night were generated and new password implemented in all HV applications. Richard. 

	[SM18 OP] 15/02/2005 11:27:07

	Could not launch test from Tema. PLCCU was set to manual instead of TEMA??? Richard. 

	 

	[SM18 OP] 15/02/2005 11:23:58

	Potaim card replaced by Richard and problem solved. 

	[SM18 OP] 15/02/2005 10:03:19

	Trigger on Potaim card of B2 being attended by Richard. 

	 

	[SM18 OP] 15/02/2005 00:15:51

	HV report cannot be generated due to change of sm18 password-Richard will take necesseray action tuesday morning 


	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _15_A


	
	Liquefactor stopped.


	ACR
	
	 4h hours of stop

	Tuesday _15_A


	TBD
	Problem in generation of HV test reports on SSS003 and SSS74


	AB/CO
	HV
	Template file was corrupted. Solved by Marco M.

	Tuesday _15_N


	TBA2


	IAP @ cold. Dc offset errors in LQA compensation. Cannot be adjusted.

	MTM
	IAP
	Richard advised to proceed. No compensation needed for the LQA the moment.

	Tuesday _15_N


	TBE2
	 MBBL2146 taken away by ROCLA before signing for stripping in paper or electronically.
	ICS
	
	 

	Wednesday_16_M


	TBA1
	CDW1--HV test report is not loaded into HV_MTF folder. But test was successful. 


	MTM
	HV
	File generated but wrong folder. Magnet is in the first run not in the third one. Re-direction of HV and Quench files performed by Richard Friday.



	Wednesday_16_N


	
	Opening/closing of water valves could not be managed from PLCCU in TEMA mode.
impossible to launch the power application from TEMA,
	AB/CO?
	
	This mode does not work. In this bench. Opening water valve with the PLC in local mode.

Test launched manually. 

	Wednesday_16_N


	TBD1
	PC fault in RMD when launching the SPA test at 1000 A.
	MTM
	
	Richard called the AB/PO first line. Second line. Key of the polarity switch not in right position.

	Thursday_17_M


	TBD1
	Not able to power CD correct in SSS003
	MTM
	
	Giorgio suspected a problem in the clamp there is NC opened for Res.Measurements (CQW0) also. The V Clamp D signal was by-passed to solve the problem temporarily.

	Friday_18_M


	TBB1


	SSS108: NC detected during HV test  (Orbit  dipole/ground) 
Resistance values for Orbit dipole outside the specs.
	
	
	Investigation by A.Siemko and Richard : caused by the coating of voltage taps wires being damaged. 
Connection for correctors not OK: Richard advised to connect the P16 connector.

	Wednesday_18_M


	TBA1
	MM1:Over-range in sector 22-23 for cmp.
	MTM
	
	Discussed with Maryline and Stephane. Go on with LHC cycle. If Over-range persists, Richard will reduce the  gain in amplifiers

	Wednesday_18_N


	TBA1
	PTE (diode test):  voltage signals have spikes (warning, D1,D2~11V) and MIITS just over the limit (24.2 MA2s). 
	MTM
	
	Richard called. He advised to continue.

	Wednesday_18_N


	TBC,TBE

TBF1
	No sufficient He liquid capacity to cool down all  the magnets: 

During PT8 De-Excitation, SPA occurred probably due to low level of LHe (As per Cryo Operator).
	ACR
	
	Put the 2 magnets in TBC and TBE in stand by.

Test re-launched later.

	Saturday_19_A


	TBF1
	Potaim Card Trigger from D2 U-L and Vtotal . 
	MTM
	
	Richard called. He adjusted offset of card 1 and changed card 7

	Saturday_19_A


	TBF1
	The two  MOBIC scanners are not working . 

One scanner was re-configured. 
	MTM
	
	Information entered manually for SSS073 and MB1080.

	Sunday_20_M


	TBF1
	Discharged of quench heater during switch on power supply. 
	MTM
	
	A connector was not properly connected.

	Sunday_20_M


	TBF1
	During warm up after the break of the vacuum from cryo, the MRB is without contact of the "interlock support MRB". 
	ACR
	
	Richard has been informed.

	Sunday_20_M

Sunday _20_N


	TBE2

TBA
	The flashing box high current is not working.

The HV rack XMT.MY95 has a connector for passive busbar broken
HV on 1186: C11 connector broken! No spare connectors!
	MTM
	
	

	Sunday_20_N


	TBF1
	Two quench heaters are not getting discharged while conducting PT5.1 on SSS110. AQA says quench not dectected.. 
	MTM
	AQA
	Test repeated 3 times, the third time at 2000 A. Quench not detected?

NC had to be open.
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